Author's response to reviews

Title: Employing crisis postcards with case management in Kaohsiung, Taiwan: 6-month outcomes of a randomised controlled trial for suicide attempters

Authors:

Wei-Jen Chen (chen8022@yahoo.com.tw)
Chi-Kung Ho (eight@kcg.gov.tw)
Shi-Sen Shyu (f50911@seed.net.tw)
Cheng-Chung Chen (ccchen@kcg.gov.tw)
Guei-Ging Lin (t1921118@yahoo.com.tw)
Yun-Ju Fang (f50911@kcg.gov.tw)
Pei-Jane Yu (jane.lin2223@msa.hinet.net)
Frank Huang-Chih Chou (f50911.tw@yahoo.com.tw)

Version: 2 Date: 31 May 2012

Author's response to reviews:

Dear Editor:

The manuscripts entitled “Employing crisis postcards with case management in Kaohsiung, Taiwan: 6-month outcomes of a randomised controlled trial for suicide attempters” submitted to the journal, BMC psychiatry, for possible publication.

We confirm that neither the material, nor any part of it, is under consideration for publication or has already been published elsewhere in any language or any form.

We are readily prepared to make any extent of concessions to increase the likelihood of publication of our work in the Journal. Your earliest response would be highly appreciated.

There are 3,664 words (including abstract, table and reference) in the manuscript and no commercial or financial involvements of a conflict of interest in connection with the submitted article.

Sincerely

Dr. Frank Huang-Chih Chou 2012/5/31